Author's response to reviews

Title: Association study of genetic variants of pro-inflammatory chemokine and cytokine genes in systemic lupus erythematosus

Authors:

Elena E Sanchez (elena@ipb.csic.es)
Jose Mario JM Sabio (masabio@terra.es)
Jose Luis JL Callejas (JLCALLEJA@telefonica.net)
Enrique E De Ramon (ederamont@telefonica.net)
Rosa R Garcia-Portales (rosagaport@hotmail.com)
Francisco J FJ Garcia-Hernandez (fjgarcia hernandez@hotmail.com)
Juan JJA Jimenez-Alonso (juan jimenez@juntadeandalucia.es)
Maria Francisca MF Gonzalez-Escribano (mariaf.gonzalez.sspa@juntadeandalucia.es)
Javier J Martin (martin@ipb.csic.es)
Bobby P BP Koeleman (b.p.c.koeleman@med.uu.nl)

Version: 8 Date: 18 May 2006

Author's response to reviews: see over
Dear Editor,

We are pleased to send you a new revised version of our manuscript entitled “Association study of genetic variants of pro-inflammatory chemokine and cytokine genes in systemic lupus erythematosus”. The manuscript has modified taking into account the format required. In the following paragraphs we comment the changes that we have included in this new reviewed version of the manuscript.

Title - We removed all unnecessary capital letter.

E-mail addresses - We changed the email addresses of FJGH and JJA

Section headings – We removed all unnecessary capital letters.

Authors’ contribution - We formatted the Authors' contributions section into a single paragraph.